Author's response to reviews

Title: Magnitude of Potentially Inappropriate Prescribing in Germany Among Older Patients with Generalized Anxiety Disorder

Authors:

Ariel Berger (aberger@pai2.com)
Marko Mychaskiw (Marko.Mychaskiw@pfizer.com)
Ellen Dukes (Ellen.Dukes@pfizer.com)
John Edelsberg (edelsberg@pai2.com)
Gerry Oster (goster@pai2.com)

Version: 3 Date: 20 July 2009

Author's response to reviews: see over
To whom it may concern:

On behalf of my colleagues, I wish to again thank you for your interest in our manuscript. Our responses to the editors’ comments are set forth below.

**Major Revisions**

1. Title – please ensure that the title is identical in the manuscript and the submission system.

   *We have ensured that the title is identical in the manuscript and the submission system.*

2. Affiliations – Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

   *We have supplied the requested affiliations in the requested format for all authors.*

3. Additional files – Additional files must be mentioned in the text and should be referred to in the following way: ‘[see Additional file 1]’ rather than as ‘Tables’.

   *Per communication with Philip Dooner (e-mail to Jamie Abraham on July 20), we have removed all additional files from our submission. We have added a “Tables” section to the body of the manuscript following the “References” section that includes all tables.*

4. Additional files – Please add a section titled “Additional files” at the end of the manuscript (after the tables) listing the following for each file: the file format, the title of the data, and a short description of the data.

   *As noted above, we have removed all additional files from the manuscript; accordingly, no further action on this comment has been taken.*

5. References – References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example ‘text [1].

   *We have formatted the references as requested.*

**Minor Revisions**

6. ‘Title, Authors’: Please remove these subheadings from the title page.

   *We have removed these subheadings from the title page, as requested.*

7. Word-counts: Please remove them.

   *Per the request, we have removed the word count from the title page.*

8. Keywords: Please remove these.
Per the request, we have removed all key words from the title page.

9. Abstract: Please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system.

Per the request, we have ensured that the Abstract is present in the manuscript and that this version of the Abstract matches exactly what is in the submission system.

Please do not hesitate to contact me with any additional questions or requests.

Sincerely,

[Signature]